# **ORIGINAL ARTICLE**

# Bexotegrast in Patients with Idiopathic Pulmonary Fibrosis The INTEGRIS-IPF Clinical Trial

Ə Lisa Lancaster<sup>1</sup>, Vincent Cottin<sup>2,3</sup>, Murali Ramaswamy<sup>4</sup>, Wim A. Wuyts<sup>5</sup>, R. Gisli Jenkins<sup>6</sup>, Mary Beth Scholand<sup>7</sup>, Michael Kreuter<sup>8</sup>, Claudia Valenzuela<sup>9</sup>, Christopher J. Ryerson<sup>10</sup>, Jonathan Goldin<sup>11,12</sup>, Grace Hyun J. Kim<sup>11,12</sup>, Marzena Jurek<sup>13</sup>, Martin Decaris<sup>13</sup>, Annie Clark<sup>13</sup>, Scott Turner<sup>13</sup>, Chris N. Barnes<sup>13</sup>, Hardean E. Achneck<sup>13</sup>, Gregory P. Cosgrove<sup>13</sup>, Éric A. Lefebvre<sup>13</sup>, and Kevin R. Flaherty<sup>14</sup>; on behalf of the PLN-74809-IPF-202 Trial Investigators

<sup>1</sup>Division of Allergy, Pulmonary, and Critical Care Medicine, Vanderbilt University Medical Center, Nashville, Tennessee; <sup>2</sup>National Reference Centre for Rare Pulmonary Diseases (Orphalung), Louis Pradel Hospital, ERN-LUNG, Lyon, France; <sup>3</sup>Claude Bernard University Lyon 1, UMR754, INRAE, Lyon, France; <sup>4</sup>Pulmonlx, LLC, at Cone Health, Greensboro, North Carolina; <sup>5</sup>Department of Pneumology, Unit for Interstitial Lung Diseases, University Hospitals Leuven, Leuven, Belgium; <sup>6</sup>National Heart and Lung Institute, Imperial College London, London, United Kingdom; <sup>7</sup>Division of Respiratory, Critical Care, and Occupational Pulmonary Medicine, University of Utah Health, Salt Lake City, Utah; <sup>8</sup>Pneumology Department, Mainz Lung Center, Mainz University Medical Center and Marienhaus Clinic Mainz, Mainz, Germany; <sup>9</sup>ILD Unit, Pulmonology Department, Hospital University of British Columbia, Vancouver, British Columbia, Canada; <sup>11</sup>Department of Radiology, University of California, Los Angeles, California; <sup>12</sup>MedQIA LLC, Los Angeles, California; <sup>13</sup>Pliant Therapeutics, Inc., South San Francisco, California; and <sup>14</sup>Division of Pulmonary and Critical Care Medicine, University of Michigan, Ann Arbor, Michigan

ORCID ID: 0000-0002-5591-0955 (V.C.).

## Abstract

**Rationale:** Idiopathic pulmonary fibrosis (IPF) is a rare and progressive disease that causes progressive cough, exertional dyspnea, impaired quality of life, and death.

**Objectives:** Bexotegrast (PLN-74809) is an oral, oncedaily, investigational drug in development for the treatment of IPF.

**Methods:** This Phase-2a multicenter, clinical trial randomized participants with IPF to receive, orally and once daily, bexotegrast at 40 mg, 80 mg, 160 mg, or 320 mg, or placebo, with or without background IPF therapy (pirfenidone or nintedanib), in an approximately 3:1 ratio in each bexotegrast dose cohort, for at least 12 weeks. The primary endpoint was incidence of treatment-emergent adverse events (TEAEs). Exploratory efficacy endpoints included change from baseline in FVC, quantitative lung fibrosis (QLF) extent (%), and changes from baseline in fibrosis-related biomarkers.

**Measurements and Main Results:** Bexotegrast was well tolerated, with similar rates of TEAEs in the pooled bexotegrast and placebo groups (62/89 [69.7%] and 21/31 [67.7%], respectively). Diarrhea was the most common TEAE; most participants with diarrhea also received nintedanib. Participants who were treated with bexotegrast experienced a reduction in FVC decline over 12 weeks compared with those who received placebo, with or without background therapy. A dose-dependent antifibrotic effect of bexotegrast was observed with QLF imaging, and a decrease in fibrosis-associated biomarkers was observed with bexotegrast versus placebo.

**Conclusions:** Bexotegrast demonstrated a favorable safety and tolerability profile, up to 12 weeks for the doses studied. Exploratory analyses suggest an antifibrotic effect according to FVC, QLF imaging, and circulating levels of fibrosis biomarkers.

Clinical trial registered with www.clinicaltrials.gov (NCT04396756).

Keywords: safety; efficacy; IPF; fibrotic disease

(Received in original form March 25, 2024; accepted in final form June 4, 2024)

3This article is open access and distributed under the terms of the Creative Commons Attribution Non-Commercial No Derivatives License 4.0. For commercial usage and reprints, please e-mail Diane Gern (dgern@thoracic.org).

A complete list of PLN-74809-IPF-202 Trial Investigators may be found before the beginning of the References.

Supported by Pliant Therapeutics, Inc.

Author Contributions: All authors contributed to, reviewed, and approved the final draft of the paper. All authors had full access to all the data in the study and had final responsibility for the decision to submit for publication. Concept and design: É.A.L. Acquisition, analysis, or interpretation of data: É.A.L., A.C., and C.N.B. Drafting of manuscript: all authors. Critical revision of the manuscript for important intellectual content: all authors. Statistical analysis: C.N.B.

Am J Respir Crit Care Med Vol 210, Iss 4, pp 424-434, Aug 15, 2024

Copyright © 2024 by the American Thoracic Society

Originally Published in Press as DOI: 10.1164/rccm.202403-0636OC on June 6, 2024 Internet address: www.atsiournals.org

## At a Glance Commentary

Scientific Knowledge on the

**Subject:** The currently approved agents for idiopathic pulmonary fibrosis (IPF), pirfenidone and nintedanib, are therapeutic options for patients with IPF; however, neither halt the progression of IPF nor consistently improve quality of life, and both have long-term tolerability issues. Bexotegrast (PLN-74809) is an oral, once-daily investigational drug that blocks the activation of transforming growth factor  $\beta$  (TGF- $\beta$ ) by preventing integrins  $\alpha_v \beta_6$  and  $\alpha_v \beta_1$  from binding to the arginine-glycineaspartate sequence of latent TGF- $\beta$ , thereby preventing TGF- $\beta$  from binding to its receptors and activating profibrogenic signaling pathways.

## What This Study Adds to the

**Field:** The results of the INTEGRIS-IPF study suggest that bexotegrast is well tolerated in patients with IPF, and exploratory analyses suggest an antifibrotic effect based on FVC, quantitative lung fibrosis imaging, and circulating levels of ITGB6 and PRO-C3, indicating the need for further studies of bexotegrast for the treatment of IPF.

Idiopathic pulmonary fibrosis (IPF) is a rare and progressive disease characterized by altered lung structure and function due to excessive extracellular matrix deposition, with resulting symptoms of progressive cough, exertional dyspnea, and impaired quality of life (1–7). Median life expectancy for untreated patients with IPF is approximately 3–4 years (2, 8, 9).

The currently approved agents for IPF, pirfenidone and nintedanib (6), represent an important therapeutic option for patients with IPF, reducing the decline of lung function and potentially reducing the time to acute exacerbations and improving survival if durable therapy is maintained (10-17). However, neither drug halts the progression of IPF nor consistently improves quality of life (10, 12, 13, 18). Both drugs have long-term tolerability issues; pirfenidone is associated with nausea and photosensitivity (12, 14, 19, 20), and nintedanib is associated with diarrhea (10, 21). Consequently, there remains an unmet need for more effective and better tolerated treatments for patients with IPF.

Transforming growth factor  $\beta$  (TGF- $\beta$ ) activation by  $\alpha_v\beta_6$  and  $\alpha_v\beta_1$  integrins is a key driver of fibrosis in the lungs (1, 22, 23).  $\alpha_v\beta_6$  and  $\alpha_v\beta_1$  integrins, expressed by lung epithelial cells, fibroblasts myofibroblasts, respectively, are both upregulated in patients with IPF (1, 23). Elevated circulating levels of integrin  $\alpha_v\beta_6$  are associated with decreased survival in patients with interstitial lung disease (24) and, specifically, IPF (25).

Bexotegrast (PLN-74809) is an oral, once-daily investigational drug in development for the treatment of IPF. Bexotegrast blocks the activation of TGF- $\beta$  by preventing  $\alpha_{\nu}\beta_{6}$  and  $\alpha_{\nu}\beta_{1}$  binding to the arginine-glycine-aspartate sequence of latent TGF- $\beta$ , thereby preventing TGF- $\beta$  from binding to its receptors and activating profibrogenic signaling pathways (1).

INTEGRIS-IPF was a Phase-2a, multicenter clinical trial evaluating the safety, tolerability, and pharmacokinetics of once-daily oral bexotegrast with or without background therapy (pirfenidone or nintedanib) in patients with IPF. The efficacy of bexotegrast in terms of FVC, quantitative lung fibrosis (QLF), and biomarkers of fibrosis was also explored.

Some of the results of these studies have been previously reported in the form of abstracts (26, 27).

## Methods

### **Study Population**

The study enrolled participants with a diagnosis of IPF according to the 2018 guidelines of the American Thoracic Society, European Respiratory Society, Japanese Respiratory Society, and Asociación Latinoamericana de Tórax (28); FVC percent predicted ≥45%; hemoglobin-adjusted  $D_{L_{CO}} \ge 30\%$ ; and a life expectancy of 6 months or longer. Background therapy for IPF with pirfenidone or nintedanib was permitted, provided that these drugs had been administered at a stable dose for 3 months or longer before screening. Key exclusion criteria included receiving any nonapproved agent intended for the treatment of pulmonary fibrosis or IPF and an FEV1/FVC ratio of <0.7. (For additional inclusion and exclusion criteria, see the online supplement.)

## **Study Design**

INTEGRIS-IPF (PLN-74809-IPF-202; ClinicalTrials.gov: NCT04396756) was a Phase-2a, multicenter, randomized, doubleblind, dose-ranging, placebo-controlled study that was conducted across 39 sites in North America, Europe, Australia, and New Zealand. This study assessed the effects of treatment with bexotegrast or placebo for at least 12 weeks in patients with IPF (Figure 1). The last study visit took place approximately 14 days after the last dose of the study drug.

Participants were randomized to 80, 160, or placebo in that cohort (40 mg or placebo; 80 mg, 160 mg or placebo; and

Data sharing statement: Clinical study data access for research use: Pliant Therapeutics, Inc. ("Pliant") understands and acknowledges the need to share clinical study data with the research community in an open and transparent manner. In furtherance of its research efforts, a member of the scientific community may request aggregated deidentified clinical data collected during a clinical study after its public disclosure by Pliant and filing of any related intellectual property protection. To the extent that Pliant has any additional supporting documentation or summary data, it may, at its discretion, also make such information available. Pliant will take into consideration any reasonable request that it receives pertaining to clinical data that have been accepted and published by a journal. Before receipt of clinical study data, Pliant and the requesting institution shall enter into an agreement that takes into consideration applicable data privacy laws and the use of the clinical study data for research purposes only. All requests for access to clinical study data must be submitted in writing to scientificcommunications@pliantrx.com.

Correspondence and requests for reprints should be addressed to Lisa Lancaster, M.D., Division of Allergy, Pulmonary, and Critical Care Medicine, Vanderbilt University Medical Center, 1161 21st Avenue S, Nashville, TN 37232. E-mail: lisa.lancaster@vumc.org.

This article has a related editorial.

A data supplement for this article is available via the Supplements tab at the top of the online article. Please also see the supplement for a plain-language summary infographic that accompanies this article.

## **ORIGINAL ARTICLE**



**Figure 1.** Study design. \*One participant, randomized to placebo, received 320 mg bexotegrast for approximately 1 week because of incorrect study-drug dispensation. No adverse events were reported for this participant. They are included in the denominator of the placebo and 320-mg bexotegrast group for the safety analyses. This figure represents the intent-to-treat population. Y/N = yes/no.

320 mg or placebo) (Figure 1). Randomization was conducted using an interactive voice-response system, and participants were stratified according to whether they were receiving or not receiving background therapy (pirfenidone or nintedanib). Participants, investigators, and those involved in the trial conduct were blinded to the trial treatment assignments.

The study was conducted in accordance with the Declaration of Helsinki, the study protocol, and International Council on Harmonisation Good Clinical Practice regulations. The study was approved by the respective Institutional Review Board, Independent Ethics Committee, or Research Ethics Board. All participants provided written informed consent.

#### **Study Assessments**

Study assessments included safety laboratory assessments, vital signs, electrocardiography, adverse events and concomitant medications, pharmacokinetic sampling, FVC (milliliters and percent predicted), FEV<sub>1</sub>, QLF extent, and the assessment of fibrosis-related biomarkers and patient-reported cough severity.

The primary endpoint was the incidence of treatment-emergent adverse events (TEAEs). Pharmacokinetic endpoints included predicted total and unbound plasma bexotegrast maximum concentration and area under the concentration time curve from Time 0 to 24 hours postdose on Days 1, 28, and 84.

Exploratory efficacy endpoints included absolute change from baseline to 12 weeks in FVC and were assessed using spirometry with central quality control. QLF extent (%) was assessed by high-resolution computed tomography (29), with central quality control, prebaseline and at Week 12. We also conducted analyses of changes from baseline to Week 12 in fibrosis-related biomarkers, serum type III collagen synthesis neoepitope (PRO-C3), and plasma integrin β6 (ITGB6).

#### **Statistical Analysis**

A sample size of approximately 21 participants per treatment group was expected to provide a meaningful evaluation of safety, tolerability, and pharmacokinetics of bexotegrast in patients with IPF (for additional details, *see* the online supplement). No power calculations were conducted, as the efficacy and biomarker assessments were exploratory in nature.

All participants who received bexotegrast or placebo were included in the safety analyses. Safety data, which were based on TEAEs, are presented by treatment group, pooled across bexotegrast doses, and included events that occurred during treatment through a 2-week follow-up period. The intent-to-treat (ITT) population included all randomized participants and the modified ITT population excluded one participant who met post hoc statistical outlier criteria, using Grubb's outlier test coupled with the clinical implausibility of the value being representative of the participant's true clinical status. FVC change from baseline was evaluated using a mixed model for repeated measures (MMRM) analysis with treatment group, visit, baseline FVC value, background therapy use (yes/no), and Treatment  $\times$  Visit interaction as terms in the model with an unstructured covariance structure: Preplanned separate subgroup

analyses were performed both for participants who received a background therapy and for those who did not. Change from baseline in biomarkers was analyzed using a similar MMRM analysis without adjustment for background therapy use. Missing data were not imputed for the MMRM analyses of FVC and biomarkers under the assumption that they were missing at random. As this was an exploratory study, no adjustment for multiplicity was planned.

For the quantitative imaging, the per-computed tomography protocol population included the subset of the ITT population with scans at baseline and Week 12 that met technical evaluability criteria per the high-resolution computed tomography imaging protocol (e.g., participants with paired examinations that met technical specifications for quantitative assessment). QLF extent is reported using observed means, SEs, and proportions.

## Results

## **Study Participants**

In total, 119 participants met the eligibility criteria and were randomized between September 2020 and June 2022 to receive, orally and once daily, bexotegrast at 40 mg (n = 22), 80 mg (n = 23), 160 mg (n = 22), or 320 mg (n = 21) or placebo (n = 31) (Figure 2). One participant, randomized to placebo, received 320 mg bexotegrast for approximately 1 week because of incorrect study-drug dispensation. This participant was included in the denominator of both treatment groups for baseline demographics and



**Figure 2.** Participant disposition. \*Adverse event (n=3), withdrawal of consent (n=1), physician decision (n=1). <sup>+</sup>Adverse event (n=1), withdrawal of consent (n=2), physician decision (n=1). <sup>+</sup>Eighty-eight participants were randomized to receive bexotegrast; however, one participant, randomized to placebo, received both placebo and 320 mg bexotegrast for approximately 1 week because of incorrect study-drug dispensation. They are included in the denominator of the placebo and 320-mg bexotegrast groups for the safety analyses.

safety analyses; however, this participant was only included in the placebo group for the efficacy analysis. The baseline demographics and disease characteristics are summarized in Table 1. The majority of participants were male, and the mean ages were 71.4 years (SD = 6.61) for bexotegrasttreated participants and 72.1 years (SD = 6.20) for placebo-treated participants. Approximately 80% of participants were receiving background therapy. Baseline FVC volume and percent predicted were comparable between treatment groups. Overall, 92% of participants completed the 12-week treatment period. A total of 9 participants prematurely discontinued treatment—5 (5.7%) in the pooled bexotegrast group and 4 (12.9%) in the placebo group-primarily because of adverse events and withdrawal of consent.

# Safety and Tolerability through Week 12

Overall, bexotegrast demonstrated a favorable safety and tolerability profile over 12 weeks of treatment (Tables 2, 3, E1, and E2 in the online supplement). The majority of TEAEs were mild or moderate in severity. The most common TEAEs are presented in Table 2. Diarrhea was the most common TEAE, occurring in 15 participants (16.9%)

in the bexotegrast (pooled) group and 3 participants (9.7%) in the placebo group. Of the participants reporting diarrhea, 13/15 (86.7%) participants in the bexotegrast group and 1/3 (33.3%) participants in the placebo group were receiving nintedanib background therapy; one participant (6.7%) with diarrhea in the bexotegrast group was receiving pirfenidone (Table 3). The remaining participant with diarrhea who received bexotegrast monotherapy had preexisting ulcerative colitis. Most events of diarrhea (14/15; 93.3%) were mild to moderate in severity; one participant receiving bexotegrast and pirfenidone had Grade 3 diarrhea. Four participants interrupted bexotegrast treatment because of intercurrent coronavirus disease (COVID-19) (n = 1), diarrhea (n = 2), and ileus and acute kidney injury (n = 1). Two participants discontinued bexotegrast because of mild diarrhea (n = 1)receiving background nintedanib; n = 1receiving no background therapy and having preexisting ulcerative colitis). Dose reduction of bexotegrast was not allowed per protocol. No serious adverse events were assessed as being related to the study drug.

TEAEs of IPF/pulmonary fibrosis were reported in 2.3% of participants in the bexotegrast group and 9.7% of participants in the placebo group. Only one of these events was reported as an acute exacerbation of IPF in a participant who had completed 12 weeks of treatment with 160 mg bexotegrast and experienced the event 11 days after the last dose. The event was not considered to be drug related and resolved the following day after treatment in hospital with corticosteroids and antibiotics.

One participant with gender-agephysiology Stage III IPF and preexisting coronary artery disease and chronic refractory atrial fibrillation in the 320-mg bexotegrast group experienced a serious and fatal adverse event of acute respiratory failure after elective atrioventricular node ablation.

There were no notable changes in laboratory parameters, vital signs, physical examination findings, or electrocardiography findings associated with the study drug. Overall, bexotegrast was well tolerated when used in combination with IPF background therapy or when used as monotherapy (Tables 3, E1, and E2).

## Pharmacokinetics

Predicted total and unbound exposures (maximum concentration and area under the concentration time curve from Time 0 to 24 hours postdose) of bexotegrast in participants with IPF increased approximately

Table 1. Baseline Demographics and Disease Characteristics

			Bexotegrast			
Demographics	$40 \mathrm{mg}  (n=22)$	80 mg ( <i>n</i> =23)	160 mg ( <i>n</i> =22)	320 mg ( <i>n</i> = 22)	Pooled ( <i>n</i> = 89)	Pooled (n = 31)
Male sex, $n$ (%) Female sex, $n$ (%) Age, yr, mean (SD) Race, $n$ (%) White White Saian Other/unknown/unreported Weight, kg, mean (SD) BMI, kg/m <sup>*</sup> , mean (SD) BMI, kg/m <sup>*</sup> , mean (SD) Dime since IPF diagnosis, mo, mean (SD) BMI, kg/m <sup>*</sup> , mean (SD) Time since IPF diagnosis, mo, mean (SD) None None Nintedanib Nintedanib Pirfenidone Duration of background therapy at randone Duration of background therapy at randona (SD) FVC, mI, mean (SD) FVC percent predicted value, mean (SD) Stage I Stage I Stage II Stage II	18 (81.8) 4 (18.2) 69.2 (7.11) 22 (100.0) 0 (0.0) 86.09 (18.22) 27.67 (4.21) 22.2 (12.44) 17 (77.3) 5 (22.7) 19.47 (11.53) 2.930.63 (789.73) 73.47 (14.37) 57.2 (14.74) 11 (50.0) 1 (45.5) 1 (4.5) 1 (4.5)	19 (82.6) 4 (17.4) 74.2 (4.70) 2 ( (91.3) 2 1 ( 91.3) 1 (4.3) 1 (4.3) 1 (4.3) 1 (4.3) 2 85.89 (14.95) 2 85.8 (5.79) 2 85.6 (17.08) 1 9 (82.6) 3 (15.90 (859.05) 8 (3.34 (14.61) 5 1.8 (14.67) 5 1.8 (14.67) 5 1.8 (14.67) 5 1.8 (14.67) 5 1.8 (14.67) 5 1.9 (16.52) 0 (0.0)	16 (72.7) 6 (27.3) 71.5 (6.63) 22 (100.0) 22 (100.0) 0 (0.0) 85.37 (13.51) 29.28 (4.66) 7 (31.8) 19 (86.4) 3 (13.6) 7 (31.8) 12 (54.5) 29.84, 7 (31.8) 28.47,10 (703.84) 7 (31.8) 28.47,10 (703.84) 78.18 (15.55) 48.6 (15.11) 2,847,10 (703.84) 78.18 (15.55) 48.6 (15.11) 2,847,10 (703.84) 78.18 (15.55) 48.6 (15.11) 2,847,10 (703.84) 7 (31.8) 2,847,10 (703.84) 7 (31.8) 2,847,10 (703.84) 7 (31.8) 2,847,10 (703.84) 7 (31.8) 2,910 2,911 2 (31.1) 2	21 (95.5) 70.5 (7.14) 21 (95.5) 21 (95.5) 21 (95.5) 21 (95.5) 21 (95.5) 21 (4.5) 88.51 (14.5) 34.4 (28.97) 14 (18.2) 10 (45.5) 34.4 (28.97) 18 (81.8) 34.4 (28.97) 10 (45.5) 33,192.00 (678.39) 77.48 (15.83) 77.48 (15.83) 77.48 (12.97) <sup>†</sup> 47.6 (12.97) <sup>†</sup> 23.9 (21.16) 23.18 (12.97) <sup>†</sup> 23.5 (2.11) 23.5 (2.11) 23.5 (2.11) 23.5 (2.11) 23.5 (2.11) 23.5 (2.11) 23.5 (2.11)	$\begin{array}{c} 74 \ (83.1) \\ 15 \ (16.9) \\ 71.4 \ (6.61) \\ 86 \ (6.61) \\ 86 \ (6.6) \\ 86 \ (6.6) \\ 11.1 \\ 28.40 \ (4.70) \\ 28.40 \ (4.70) \\ 28.2 \ (19.12) \\ 73 \ (82.0) \\ 16 \ (18.0) \\ 38 \ (42.7) \\ 58 \ (42.7) \ (42.7) \ (42.7) \ (42.7) \ (42.7) \ (42.7) \ (42.7) \ (42.7) \ (42.7) \ (42.7) \ (42.7) \ ($	27 (87.1) 4 (12.9) 72.1 (6.20) 30 (96.8) 1 (3.2) 0 (0.0) 83.95 (11.41)* 24 (77.4) 7 (22.6) 13 (41.9) 11 (35.5) 22.55 (17.85) 3,083.62 (783.34) 78.05 (17.16) 50.1 (15.23) 50.1 (15.23) 3 (9.7) 3 (9.7)
*Data available for 29 participants. <sup>†</sup> Data available for 21 participants. <sup>‡</sup> Data available for 88 participants.			_			

**ORIGINAL ARTICLE** 

## Table 2. Nature and Frequency of TEAEs Over 12 Weeks of Treatment

		Participants, <i>n</i> (%)					
		Bexotegrast					
Nature of TEAE	40 mg (n = 22)	80 mg (n = 23)	160 mg ( <i>n</i> = 22)	320 mg ( <i>n</i> = 22)	Pooled ( <i>n</i> = 89)	Placebo ( <i>n</i> = 31)	
TEAE TEAE related to study drug TEAE ≥Grade 3 TEAE ≥Grade 3 related to study drug SAE SAE related to study drug TEAE leading to interruption of study drug TEAE leading to withdrawal of study drug TEAE leading to early termination from the study TEAE leading to death TEAE leading to death TEAE soccurring in ≥5% of participants in any group (any causality)	$\begin{array}{c} 16 & (72.7) \\ 4 & (18.2) \\ 2 & (9.1) \\ 0 & (0.0) \\ 1 & (4.5) \\ 0 & (0.0) \\ 0 & (0.0) \\ 0 & (0.0) \\ 0 & (0.0) \\ 0 & (0.0) \\ 0 & (0.0) \end{array}$	$\begin{array}{c} 15 \ (65.2) \\ 7 \ (30.4) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 0 \ (0.0) \end{array}$	$\begin{array}{c} 14 \ (63.6) \\ 4 \ (18.2) \\ 2 \ (9.1) \\ 1 \ (4.5)^{\star} \\ 2 \ (9.1) \\ 0 \ (0.0) \\ 1 \ (4.5)^{\dagger} \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 0 \ (0.0) \end{array}$	$\begin{array}{c} 17 \ (77.3) \\ 4 \ (18.2) \\ 2 \ (9.1) \\ 0 \ (0.0) \\ 1 \ (4.5) \\ 0 \ (0.0) \\ 1 \ (4.5)^{\ddagger} \\ 3 \ (13.6)^{\ddagger,\$,  } \\ 3 \ (13.6)^{\ddagger,\$,  } \\ 1 \ (4.5)^{\$} \end{array}$	62 (69.7) 19 (21.3) 6 (6.7) 1 (1.1) 4 (4.5) 0 (0.0) 2 (2.2) 3 (3.4) 3 (3.4) 1 (1.1)	21 (67.7) 10 (32.3) 2 (6.5) 0 (0.0) 3 (9.7) 0 (0.0) 0 (0.0) 3 (9.7) 2 (6.5) 0 (0.0)	
Diarrhea Diarrhea Fatigue Nausea Dyspnea Atrioventricular block, first degree Vomiting COVID-19 Cough Headache IPF/pulmonary fibrosis Hyperkalemia Skin abrasion	$\begin{array}{c} 2 & (9.1) \\ 2 & (9.1) \\ 1 & (4.5) \\ 0 & (0.0) \\ 0 & (0.0) \\ 0 & (0.0) \\ 1 & (4.5) \\ 0 & (0.0) \\ 1 & (4.5) \\ 0 & (0.0) \\ 0 & (0.0) \end{array}$	$\begin{array}{c} 5 (21.7) \\ 2 (8.7) \\ 1 (4.3) \\ 1 (4.3) \\ 0 (0.0) \\ 0 (0.0) \\ 0 (0.0) \\ 1 (4.3) \\ 0 (0.0) \\ 0 (0.0) \\ 0 (0.0) \\ 0 (0.0) \\ 0 (0.0) \end{array}$	$\begin{array}{c} 5 (22.7) \\ 1 (4.5) \\ 2 (9.1) \\ 2 (9.1) \\ 0 (0.0) \\ 2 (9.1) \\ 2 (9.1) \\ 1 (4.5) \\ 0 (0.0) \\ 1 (4.5) \\ 0 (0.0) \\ 0 (0.0) \end{array}$	$\begin{array}{c} 3 \ (13.6) \\ 2 \ (9.1) \\ 0 \ (0.0) \\ 1 \ (4.5) \\ 2 \ (9.1) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 1 \ (4.5) \\ 1 \ (4.5) \\ 1 \ (4.5) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ 0 \ (0.0) \\ \end{array}$	$\begin{array}{c} 15 \ (16.9) \\ 7 \ (7.9) \\ 4 \ (4.5) \\ 2 \ (2.2) \\ 2 \ (2.2) \\ 2 \ (2.2) \\ 2 \ (2.2) \\ 4 \ (4.5) \\ 1 \ (1.1) \\ 2 \ (2.3) \\ 0 \ (0.0) \\ 0 \ (0.0) \end{array}$	3 (9.7) 2 (6.5) 2 (6.5) 0 (0.0) 2 (6.5) 0 (0.0) 2 (6.5) 2 (6.5) 3 (9.7) 2 (6.5) 2 (6.5) 2 (6.5)	

*Definition of abbreviations*: COVID-19 = coronavirus disease; GAP = gender-age-physiology; IPF = idiopathic pulmonary fibrosis; SAE = serious adverse event; TEAE = treatment-emergent adverse event.

\*Grade 3 event of diarrhea in a participant taking pirfenidone.

<sup>†</sup>Participant had a TEAE of COVID-19 that led to the interruption of the study drug.

<sup>‡</sup>Abdominal pain/diarrhea in a participant with preexisting ulcerative colitis.

<sup>§</sup>Acute respiratory failure in a GAP Stage III participant with preexisting coronary artery disease and atrial fibrillation after elective atrioventricular node ablation.

<sup>II</sup>Diarrhea in a participant with concomitant use of nintedanib.

proportionally with dose (*see* Tables E3 and E4).

## Efficacy (Exploratory Endpoints)

Bexotegrast treatment resulted in reduced FVC decline over 12 weeks compared with placebo. In the modified ITT population, in which 81% of the participants received background therapy, changes in FVC from baseline to Week 12 were -46.1 ml (SE = 39.64), 25.6 ml (SE = 38.57), -25.6 ml (SE = 40.49), and 29.4 ml (SE = 43.35), respectively, for participants in the groups receiving 40, 80, 160, and 320 mg bexotegrast and -3.6 (SE = 21.97) for the pooled bexotegrast treatment cohort, compared with -110.5 ml (SE = 36.38) for participants in the placebo group (Figures 3 and E1). At Week 12, changes in FVC with bexotegrast were statistically significant compared with placebo in the groups who received 80 mg (difference: 136.1 ml; P = 0.0094) and 320 mg

bexotegrast (difference: 139.9 ml; P = 0.0124). The treatment effect of bexotegrast was observed in participants receiving or not receiving background therapy for IPF (*see* Figure E2). In the extension study, of patients taking 320 mg bexotegrast who had an increase in FVC at Week 12 (n = 10), 88.9% maintained that increase to Week 24; in the placebo group, no participants maintained their FVC increase observed at Week 12 to Week 24 (*see* Figure E3).

A dose-dependent trend of a reduction in the percentage of participants with relative and absolute decline  $\geq 10\%$  of FVC percent predicted (FVCpp) was observed over the 12-week treatment period (Figure 4).

Quantitative assessment of lung imaging demonstrated that mean percent change in QLF extent decreased dose-dependently from baseline to Week 12 with no or limited progression at 160 mg and 320 mg (Figure 5A). Circulating biomarker analysis revealed an overall dose-dependent decrease in ITGB6 relative to placebo and reductions that were significant at Week 4 for bexotegrast doses at 80, 160 and 320 mg and that continued to be significant and declined further at Week 12 (Figure 5B). For PRO-C3, there was a significant reduction, compared with placebo, for the 320-mg dose at Week 4; and a further reduction compared with placebo at Week 12 was observed for the 80- and 160-mg doses compared with Week 4 (Figure 5C).

## Discussion

The INTEGRIS-IPF study demonstrated a positive safety and tolerability profile for bexotegrast up to 12 weeks, with no dose relationship for TEAEs across the four doses studied (for a plain-language infographic, *see* the online supplement). The most common

	Participants, n (%)							
Nature of TEAE	Pooled Bexotegrast, No Background Therapy ( <i>n</i> = 16)	Pooled Bexotegrast + Nintedanib (n = 38)	Pooled Bexotegrast + Pirfenidone ( <i>n</i> = 35)	Placebo (n = 7)	Placebo + Nintedanib (n = 13)	Placebo + Pirfenidone (n = 11)		
TEAE TEAEs occurring in ≥5% of participants in any overall group (any causality)	12 (75.0)	10 (76.9)	20 (57.1)	5 (71.4)	33 (86.8)	6 (54.5)		
Diarrhea	1 (6.3)	16 (42.1)	2 (5.7)	2 (28.6)	2 (15.4)	0 (0.0)		
Fatigue	2 (12.5)	1 (2.6)	4 (11.4)	1 (14.3)	0 (0.0)	1 (9.1)		
Nausea	2 (12.5)	1 (2.6)	1 (2.9)	1 (14.3)	1 (7.7)	1 (9.1)		
Dyspnea	2 (12.5)	3 (7.9)	3 (8.6)	0 (0.0)	1 (7.7)	0 (0.0)		
Atrioventricular block, first degree	1 (6.3)	0 (0.0)	1 (2.9)	0 (0.0)	0 (0.0)	2 (18.2)		
Vomiting	0 (0.0)	2 (5.3)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)		
COVID-19	1 (6.3)	2 (5.3)	1 (2.9)	0 (0.0)	0 (0.0)	0 (0.0)		
Cough	1 (6.3)	3 (7.9)	2 (5.7)	1 (14.3)	1 (7.7)	1 (9.1)		
Headache	0 (0.0)	0 (0.0)	0 (0.0)	2 (28.6)	0 (0.0)	1 (9.1)		
IPF/pulmonary fibrosis	0 (0.0)	3 (7.9)	3 (8.6)	0 (0.0)	0 (0.0)	1 (9.1)		
Hyperkalemia	0 (0.0)	0 (0.0)	0 (0.0)	1 (14.3)	1 (7.7)	1 (9.1)		
Skin abrasion	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	1 (7.7)	1 (9.1)		

Table 3. Nature and Frequency of TEAEs With or Without Background Therapy Over 12 Weeks of Treatment

Definition of abbreviations: COVID-19 = coronavirus disease; IPF = idiopathic pulmonary fibrosis; TEAE = treatment-emergent adverse event.



**Figure 3.** Change in FVC from baseline to Week 12 (efficacy for modified intent-to-treat population). (A–D) Change in FVC from baseline through Week 12 in all participants receiving (A) 40 mg, (B) 80 mg, (C) 160 mg, and (D) 320 mg bexotegrast compared with placebo. Change from baseline was analyzed using a mixed model for repeated measures. \*P<0.05 versus placebo. \*\*P<0.01 versus placebo. LS = least squares.

## **ORIGINAL ARTICLE**



**Figure 4.** Percentage of participants with FVC (*A*) relative decline  $\geq$ 10% of predicted and (*B*) absolute decline  $\geq$ 10% of predicted at Week 12 (efficacy for modified intent-to-treat population).

adverse event reported was diarrhea, primarily observed in those taking background nintedanib. Notably, no relationship between the dose of bexotegrast and incidence of diarrhea was observed. Gastrointestinal adverse events, including diarrhea, are common adverse reactions in patients treated with nintedanib or pirfenidone (10, 12–14, 16). The present study did not identify any safety concerns for bexotegrast and confirms the positive safety profile identified in the 7 completed Phase-1 studies, in which the most common TEAEs (occurring in more than 5% of participants receiving bexotegrast) were



**Figure 5.** Exploratory endpoints. (*A*) Mean change in QLF (%) from baseline to Week 12. (*B*) Least squares (LS) mean change from baseline in plasma ITGB6 relative to placebo at Weeks 4 and 12. (*C*) LS mean change from baseline in serum PRO-C3 relative to placebo at Weeks 4 and 12. The data in (*A*) are per computed tomography protocol population (a subset of the intent-to-treat population with available high-resolution computer tomography imaging data and images) who completed the study without any major protocol violations within the prespecified time interval between screening and randomization. The data in (*B*) and (*C*) are from the pharmacodynamic analysis population. Change from baseline in biomarker concentration was analyzed using a mixed model for repeated measures. \**P*<0.05 versus placebo. \*\**P*<0.01 versus placebo. \*\**P*<0.001 versus placebo. ITGB6 = integrin β-6; NPX = Normalized Protein eXpression; PRO-C3 = type III collagen synthesis neoepitope; QLF = quantitative lung fibrosis.

headache and constipation (30). No drugrelated serious adverse events or deaths were reported in any of these studies.

Previously, treatment with the anti- $\alpha_{y}\beta_{6}$ IgG1 monoclonal antibody BG00011 (up to 8 wk of treatment in a Phase-2a study and up to 40 wk of treatment in a Phase-2b study) was shown to be associated with a significant imbalance of IPF acute exacerbations compared with placebo (31, 32). The reported increase in acute exacerbations of IPF was postulated to be a result of the proinflammatory potential of prolonged TGF- $\beta$  inhibition (32). However, no increased risk of disease progression, denoted by IPF or pulmonary fibrosis, was identified in the present study of bexotegrast over a 12-week treatment duration. In the extended treatment duration for participants who were enrolled in the 320-mg bexotegrast cohort (mean exposure, 27.2 wk), the favorable safety profile extended to at least 24 weeks and up to 40 weeks (26). Chronic toxicology studies of bexotegrast as well as prior studies with systemic inhibition of TGF-β with neutralizing antibodies or small molecule inhibitor have not suggested a proinflammatory effect in the lungs of rodents and nonhuman primates (33). In our prior investigation in healthy volunteers exposed to bexotegrast for 7 days (PLN-74809-109) (30), there were no notable differences in the total leukocyte cell count or distribution across bexotegrast treatment groups (40, 80, 160, and 320 mg) compared with placebo. Therefore, the increased risk of IPF acute exacerbation noted with BG00011 may be drug specific and not a result of prolonged TGF-β inhibition. Late-stage evaluation of the efficacy and safety of bexotegrast is underway in the 52-week BEACON-IPF trial (PLN-74809-IPF-206; ClinicalTrials.gov: NCT06097260), which will further characterize its benefit-risk assessment

Pharmacokinetic data indicated doseproportional increases in bexotegrast exposure, which were consistent with the findings of previous studies (34).

The exploratory efficacy and biomarker data presented here provide the first evidence of the potential effect of bexotegrast in participants with IPF. Previously, bexotegrast showed sustained reduction of TGF- $\beta$ signaling in BAL cells collected from healthy participants (PLN-74809-109) (35), as well as antifibrotic effects in *ex vivo* human and *in vivo* murine models of pulmonary fibrosis (1). In the present study, compared with placebo, bexotegrast was associated with reduced FVC decline and a trend toward a reduction in participants with relative and absolute decline in FVCpp  $\geq 10\%$ , which are predictors of increased mortality and disease progression (36, 37). Statistical significance was identified for the absolute change in FVC (in milliliters) at Week 12 for the 80-mg and 320-mg doses but not the 160-mg dose, which may reflect the inherent variability of FVC measurement, short treatment duration or limitations of smaller samples sizes in the individual cohorts. The treatment effect of the 160-mg dose at Week 12 is supported by the totality of the data, including an FVCpp  $\geq$ 10% (absolute and relative), mean change in QLF, cough severity as assessed by visual analog scale, and change in the prognostic biomarkers ITGB6 and PRO-C3. For these endpoints, a clear dose relationship was observed, with the 160-mg and 320-mg doses providing the best results. The placebo cohort appeared to have a large decline of -110 ml in FVC over 12 weeks, in which approximately 80% of the patients were receiving background therapy, consistent with findings observed in recent IPF trials (38). The progressive nature of decline in participants receiving background therapy in recent trials remains unexplained. Direct comparison with prior registrational trials for pirfenidone and nintedanib are limited because of the lack of consistent exposure to therapeutic doses, either because of dose reductions or dose interruptions or because of the mean dose at baseline for participants being lower than the recommended dosage in recent Phase-3 trials (38). The data from INTEGRIS-IPF were, however, similar to what was observed in placebo groups for other Phase-2 IPF trials in which there was a range in FVC decline from -60 to -101 ml (16, 39-41). Last, a post hoc analysis of placebo groups in pirfenidone clinical trials examined FVC variability at 3-month intervals and found that approximately 50% of all 3-month intervals assessed had an increase or decrease of ≥100 ml, suggesting that 12-week intervals can be variable (42). Other factors can also contribute to shifts in FVC, such as the intrinsic variability of the measure itself; patient demographics, including being male or White; patient progression being unpredictable; differences in patient background therapy; inclusion and analysis of missing data; and patient sample size (43-45). Additional studies in larger patient

populations will confirm the robustness of the treatment effect by bexotegrast.

The majority of study participants were receiving background therapy for IPF. Results suggest that adding bexotegrast to an approved IPF background therapy over a 12-week period improved prevention of FVC decline compared with pirfenidone or nintedanib plus placebo, without additive toxicity. IPF progression in participants receiving bexotegrast or placebo with background therapy was consistent with recent late-stage studies of IPF combination therapies (38, 41).

The potential additive benefit associated with bexotegrast to IPF was supported by initial radiographic evidence of reduced increases in QLF extent versus placebo. The extent of QLF correlates with measures of lung function, such as FVCpp and DL<sub>CO</sub>, in longitudinal evaluations at 6 and 12 months (46, 47). A minimally important clinical difference of 2% has been validated in natural history cohorts of IPF (48, 49), and larger studies over a longer period of time will be needed to assess the correlation of QLF and FVC with bexotegrast.

It is interesting that a longitudinal effect of therapy was observed on circulating levels of integrin  $\alpha_{v}\beta_{6}$  (ITGB6), a biomarker previously associated with disease progression (as defined by death, lung transplant, or ≥10% relative FVC decline) in patients with interstitial lung disease (24). There was a significant reduction in ITGB6 levels with bexotegrast compared with placebo, which is consistent with published evidence of the effect of TGF- $\beta$  on *ITGB6* gene expression (50). PRO-C3 is also reportedly elevated in patients with IPF and associated with progressive disease (51). Our analyses revealed that circulating levels of PRO-C3 were reduced in a dose-dependent manner in participants receiving 80, 160, and 320 mg bexotegrast compared with placebo. Additional investigation is required to validate biomarkers in IPF.

Strengths of the present study include its prospective, dose-ranging, randomized design and the inclusion of background therapies for IPF. Although interpretation of safety data is limited by the relatively short treatment period, in terms of efficacy, changes in FVC at 3 months may be a surrogate marker for mortality, disease progression, and change in FVC at 12 months (37). Another limitation of the present design is that it was not powered to adequately assess efficacy with relatively small patient numbers in each dose group.

Given the small number of participants, we should interpret with caution the observed absolute change in FVC (in milliliters) over the dose range of 80 to 320 mg. The majority of participants in this study were male and White, consistent with other registrational trials in IPF. Given the recent regulatory guidance to have clinical trials include underrepresented racial and ethnic populations, as well as the high need for effective treatments for all patients, even those with rare diseases such as IPF, future studies should include a broader representation of patients. Longer term (52-wk) treatment in the BEACON-IPF trial will assess the durability of the antifibrotic effects and safety profile, as well as the potential to provide meaningful clinical benefits, such as stabilization of

the progressive fibrosis and symptomatic deterioration, in patients with IPF.

In summary, the results of the INTEGRIS-IPF study suggest that bexotegrast is well tolerated in patients with IPF, with no observed dose-dependent relationship for AEs, and exploratory analyses suggest an antifibrotic effect according to FVC, QLF imaging, and circulating levels of ITGB6 and PRO-C3. These results support the need for further studies of bexotegrast for the treatment of IPF.

Author disclosures are available with the text of this article at www.atsjournals.org.

**Acknowledgment:** Editorial assistance was provided by Denise Kenski of Nucleus Global, and Aimee Sherlock of Alpharmaxim

Healthcare Communications and was funded by Pliant Therapeutics, Inc. The authors thank the clinical site staff, the patients who participated in this study, and all those who donate time to the evaluation of more effective treatments for IPF.

## PLN-74809-IPF-202 Trial Investigators:

David Baratz, Neil Ettinger, Daniel Layish, Matthew D. Epstein, Srihari Veeraraghavan, Jeffrey Golden, Murali Ramaswamy, Rebecca Bascom, Lisa Lancaster, Mary Beth Scholand, Amy Case, Tanzira Zaman, Alan Betensley, Danielle Antin-Ozerkis, Sydney Montessi, Evans Fernandez, Ryan Boente, Jeffrey Sager, Gary Hunninghake, Kevin Gibson, Nadim Srour, Anil Dhar, Wim Wuyts, Pascal Wielders, Marcel Veltkamp, Remy Mostard, Robert Janssen, Anton Vonk Noordegraaf, Ian Glaspole, Tamera Corte, Lutz Beckert, Benedict (Ben) Brockway, Andrew Veale, Luca Richeldi, and Sergio Harari.

#### References

- 1. Decaris ML, Schaub JR, Chen C, Cha J, Lee GG, Rexhepaj M, *et al.* Dual inhibition of  $\alpha_{v}\beta_{6}$  and  $\alpha_{v}\beta_{1}$  reduces fibrogenesis in lung tissue explants from patients with IPF. *Respir Res* 2021;22:265.
- Lederer DJ, Martinez FJ. Idiopathic pulmonary fibrosis. N Engl J Med 2018;378:1811–1823.
- Maher TM, Bendstrup E, Dron L, Langley J, Smith G, Khalid JM, et al. Global incidence and prevalence of idiopathic pulmonary fibrosis. *Respir Res* 2021;22:197.
- Booth AJ, Hadley R, Cornett AM, Dreffs AA, Matthes SA, Tsui JL, et al. Acellular normal and fibrotic human lung matrices as a culture system for in vitro investigation. Am J Respir Crit Care Med 2012;186:866–876.
- Guo H, Sun J, Zhang S, Nie Y, Zhou S, Zeng Y. Progress in understanding and treating idiopathic pulmonary fibrosis: recent insights and emerging therapies. *Front Pharmacol* 2023;14:1205948.
- Mai TH, Han LW, Hsu JC, Kamath N, Pan L. Idiopathic pulmonary fibrosis therapy development: a clinical pharmacology perspective. *Ther Adv Respir Dis* 2023;17:17534666231181537.
- Raghu G, Remy-Jardin M, Richeldi L, Thomson CC, Inoue Y, Johkoh T, et al. Idiopathic pulmonary fibrosis (an update) and progressive pulmonary fibrosis in adults: an official ATS/ERS/JRS/ALAT clinical practice guideline. *Am J Respir Crit Care Med* 2022;205: e18–e47.
- Raghu G, Chen SY, Yeh WS, Maroni B, Li Q, Lee YC, et al. Idiopathic pulmonary fibrosis in US Medicare beneficiaries aged 65 years and older: incidence, prevalence, and survival, 2001-11. *Lancet Respir Med* 2014;2:566–572.
- Strongman H, Kausar I, Maher TM. Incidence, prevalence, and survival of patients with idiopathic pulmonary fibrosis in the UK. *Adv Ther* 2018;35: 724–736.
- Lancaster L, Crestani B, Hernandez P, Inoue Y, Wachtlin D, Loaiza L, et al. Safety and survival data in patients with idiopathic pulmonary fibrosis treated with nintedanib: pooled data from six clinical trials. BMJ Open Respir Res 2019;6:e000397.
- Nathan SD, Albera C, Bradford WZ, Costabel U, Glaspole I, Glassberg MK, et al. Effect of pirfenidone on mortality: pooled analyses and metaanalyses of clinical trials in idiopathic pulmonary fibrosis. *Lancet Respir Med* 2017;5:33–41.
- King TE Jr, Bradford WZ, Castro-Bernardini S, Fagan EA, Glaspole I, Glassberg MK, et al.; ASCEND Study Group. A phase 3 trial of pirfenidone in patients with idiopathic pulmonary fibrosis. N Engl J Med 2014;370:2083–2092.
- Richeldi L, du Bois RM, Raghu G, Azuma A, Brown KK, Costabel U, et al.; INPULSIS Trial Investigators. Efficacy and safety of nintedanib in idiopathic pulmonary fibrosis. N Engl J Med 2014;370:2071–2082.

- Noble PW, Albera C, Bradford WZ, Costabel U, Glassberg MK, Kardatzke D, et al.; CAPACITY Study Group. Pirfenidone in patients with idiopathic pulmonary fibrosis (CAPACITY): two randomised trials. Lancet 2011;377:1760–1769.
- Petnak T, Lertjitbanjong P, Thongprayoon C, Moua T. Impact of antifibrotic therapy on mortality and acute exacerbation in idiopathic pulmonary fibrosis: a systematic review and meta-analysis. *Chest* 2021;160:1751–1763.
- Richeldi L, Costabel U, Selman M, Kim DS, Hansell DM, Nicholson AG, et al. Efficacy of a tyrosine kinase inhibitor in idiopathic pulmonary fibrosis. N Engl J Med 2011;365:1079–1087.
- Richeldi L, Cottin V, du Bois RM, Selman M, Kimura T, Bailes Z, et al. Nintedanib in patients with idiopathic pulmonary fibrosis: combined evidence from the TOMORROW and INPULSIS(<sup>®</sup>) trials. *Respir Med* 2016;113:74–79.
- Fisher M, Nathan SD, Hill C, Marshall J, Dejonckheere F, Thuresson PO, et al. Predicting life expectancy for pirfenidone in idiopathic pulmonary fibrosis. J Manag Care Spec Pharm 2017;23(Suppl. 3b): S17–S24.
- Costabel U, Albera C, Lancaster LH, Lin CY, Hormel P, Hulter HN, et al. An open-label study of the long-term safety of pirfenidone in patients with idiopathic pulmonary fibrosis (RECAP). *Respiration* 2017;94: 408–415.
- Cottin V, Koschel D, Günther A, Albera C, Azuma A, Sköld CM, et al. Long-term safety of pirfenidone: results of the prospective, observational PASSPORT study. ERJ Open Res 2018;4: 00084-02018.
- Crestani B, Huggins JT, Kaye M, Costabel U, Glaspole I, Ogura T, et al. Long-term safety and tolerability of nintedanib in patients with idiopathic pulmonary fibrosis: results from the open-label extension study, INPULSIS-ON. Lancet Respir Med 2019;7:60–68.

22. John AE, Graves RH, Pun KT, Vitulli G, Forty EJ, Mercer PF, *et al.* Translational pharmacology of an inhaled small molecule  $\alpha_{\nu}\beta_{6}$  integrin inhibitor for idiopathic pulmonary fibrosis. *Nat Commun* 2020;11:4659.

- 23. Saini G, Porte J, Weinreb PH, Violette SM, Wallace WA, McKeever TM, et al.  $\alpha_{\nu}\beta_{6}$  integrin may be a potential prognostic biomarker in interstitial lung disease. *Eur Respir J* 2015;46:486–494.
- Bowman WS, Newton CA, Linderholm AL, Neely ML, Pugashetti JV, Kaul B, et al. Proteomic biomarkers of progressive fibrosing interstitial lung disease: a multicentre cohort analysis. *Lancet Respir Med* 2022;10: 593–602.
- Oldham J, Huang Y, Ma SF, Lee CT, Kim J, Pugashetti JV, et al. Proteomic determinants of idiopathic pulmonary fibrosis survival. Am J Respir Crit Care Med 2024;209:1111–1120.
- 26. Wuyts W, Valenzuela C, Jenkins G, Goldin J, Kim G, Jurek M, et al. Late breaking abstract—Safety, tolerability and antifibrotic activity of

bexotegrast: phase 2a. INTEGRIS-IPF study (NCT04396756). Eur Respir J 2023;62:OA1423.

- 27. Lancaster L, Cottin V, Ramaswamy M, Goldin J, Kim G, Bellini J, et al. PLN-74809 shows favorable safety and tolerability and indicates antifibrotic activity in a phase 2a study for the treatment of idiopathic pulmonary fibrosis. Am J Respir Crit Care Med 2023;207:A2777.
- Raghu G, Remy-Jardin M, Myers JL, Richeldi L, Ryerson CJ, Lederer DJ, et al.; American Thoracic Society, European Respiratory Society, Japanese Respiratory Society, and Latin American Thoracic Society. Diagnosis of idiopathic pulmonary fibrosis. An official ATS/ERS/JRS/ ALAT clinical practice guideline. Am J Respir Crit Care Med 2018;198: e44–e68.
- Kim HG, Tashkin DP, Clements PJ, Li G, Brown MS, Elashoff R, et al. A computer-aided diagnosis system for quantitative scoring of extent of lung fibrosis in scleroderma patients. *Clin Exp Rheumatol* 2010; 28(5 Suppl. 62):S26–S35.
- 30. Cosgrove G, Park E, Bellini J, Adegbite E, Cilli F, Jurek M, *et al.* PLN-74809, a dual-selective inhibitor of  $\alpha_{\nu}\beta_{6}$  and  $\alpha_{\nu}\beta_{1}$ , is well tolerated in over 280 healthy participants. *Am J Respir Crit Care Med* 2022;205: A2437.
- 31. Raghu G, Mouded M, Chambers DC, Martinez FJ, Richeldi L, Lancaster LH, *et al.* A phase IIb randomized study of an anti- $\alpha_{\nu}\beta_{6}$  monoclonal antibody in idiopathic pulmonary fibrosis. *Am J Respir Crit Care Med* 2022;206:1128–1139.
- 32. Raghu G, Mouded M, Prasse A, Stebbins C, Zhao G, Song G, *et al.* Randomized phase IIa clinical study of an anti- $\alpha_{v}\beta_{6}$  monoclonal antibody in idiopathic pulmonary fibrosis. *Am J Respir Crit Care Med* 2022;206:1166–1168.
- 33. Mitra MS, Lancaster K, Adedeji AO, Palanisamy GS, Dave RA, Zhong F, et al. A potent pan-TGFβ neutralizing monoclonal antibody elicits cardiovascular toxicity in mice and cynomolgus monkeys. *Toxicol Sci* 2020;175:24–34.
- 34. Mooney JJ, Morris K, Jacobs S, Lefebvre E, Cosgrove G, Wong S, *et al.* PLN-74809, a dual-selective inhibitor of integrins  $\alpha_v\beta_6$  and  $\alpha_v\beta_1$ , shows dose-dependent target engagement in the lungs of patients with idiopathic pulmonary fibrosis (IPF). *Am J Respir Crit Care Med* 2022; 205:A5254.
- 35. Cosgrove G, Decaris M, Park E, Wong S, Park E, Turner S, et al. PLN-74809, an oral, dual-selective αvβ6/αvβ1 inhibitor in phase 2 clinical trials for idiopathic pulmonary fibrosis (IPF), sustainably reduces transforming growth factor-beta (TGF-β) activity in the lungs of healthy participants with once-daily dosing. Am J Respir Crit Care Med 2022; 205:A5251.
- 36. Paterniti MO, Bi Y, Rekić D, Wang Y, Karimi-Shah BA, Chowdhury BA. Acute exacerbation and decline in forced vital capacity are associated with increased mortality in idiopathic pulmonary fibrosis. *Ann Am Thorac Soc* 2017;14:1395–1402.
- 37. Khan FA, Stewart I, Moss S, Fabbri L, Robinson KA, Johnson SR, et al. Three-month FVC change: a trial endpoint for idiopathic pulmonary fibrosis based on individual participant data meta-analysis. Am J Respir Crit Care Med 2022;205:936–948.

- Maher TM, Ford P, Brown KK, Costabel U, Cottin V, Danoff SK, et al.; ISABELA 1 and 2 Investigators. Ziritaxestat, a novel autotaxin inhibitor, and lung function in idiopathic pulmonary fibrosis: the ISABELA 1 and 2 randomized clinical trials. JAMA 2023;329:1567–1578.
- 39. Maher TM, van der Aar EM, Van de Steen O, Allamassey L, Desrivot J, Dupont S, et al. Safety, tolerability, pharmacokinetics, and pharmacodynamics of GLPG1690, a novel autotaxin inhibitor, to treat idiopathic pulmonary fibrosis (FLORA): a phase 2a randomised placebo-controlled trial. *Lancet Respir Med* 2018;6:627–635.
- 40. Strambu IR, Seemayer CA, Fagard LMA, Ford PA, Van der Aa TAK, de Haas-Amatsaleh AA, et al. GLPG1205 for idiopathic pulmonary fibrosis: a phase 2 randomised placebo-controlled trial. *Eur Respir J* 2023;61: 2201794.
- Richeldi L, Azuma A, Cottin V, Hesslinger C, Stowasser S, Valenzuela C, et al.; 1305-0013 Trial Investigators. Trial of a preferential phosphodiesterase 4B inhibitor for idiopathic pulmonary fibrosis. N Engl J Med 2022;386:2178–2187.
- Nathan SD, Yang M, Morgenthien EA, Stauffer JL. FVC variability in patients with idiopathic pulmonary fibrosis and role of 6-min walk test to predict further change. *Eur Respir J* 2020;55:1902151.
- 43. Santermans E, Ford P, Kreuter M, Verbruggen N, Meyvisch P, Wuyts WA, et al. Modelling forced vital capacity in idiopathic pulmonary fibrosis: optimising trial design. Adv Ther 2019;36:3059–3070.
- 44. Thabut G, Crestani B, Porcher R, Richeldi L. Missing data in IPF trials: do not let methodological issues undermine a major therapeutic breakthrough. *Eur Respir J* 2015;46:607–614.
- 45. Neely ML, Hellkamp AS, Bender S, Todd JL, Liesching T, Luckhardt TR, et al. Lung function trajectories in patients with idiopathic pulmonary fibrosis. *Respir Res* 2023;24:209.
- 46. Kim HJ, Brown MS, Chong D, Gjertson DW, Lu P, Kim HJ, et al. Comparison of the quantitative CT imaging biomarkers of idiopathic pulmonary fibrosis at baseline and early change with an interval of 7 months. Acad Radiol 2015;22:70–80.
- 47. Kafaja S, Clements PJ, Wilhalme H, Tseng CH, Furst DE, Kim GH, et al. Reliability and minimal clinically important differences of forced vital capacity: results from the Scleroderma Lung Studies (SLS-I and SLS-II). Am J Respir Crit Care Med 2018;197:644–652.
- 48. Kim GHJ, Goldin JG, Hayes W, Oh A, Soule B, Du S. The value of imaging and clinical outcomes in a phase II clinical trial of a lysophosphatidic acid receptor antagonist in idiopathic pulmonary fibrosis. *Ther Adv Respir Dis* 2021;15:17534666211004238.
- 49. Kim GHJ, Weigt SS, Belperio JA, Brown MS, Shi Y, Lai JH, et al. Prediction of idiopathic pulmonary fibrosis progression using early quantitative changes on CT imaging for a short term of clinical 18-24month follow-ups. *Eur Radiol* 2020;30:726–734.
- Sheppard D, Cohen DS, Wang A, Busk M. Transforming growth factor beta differentially regulates expression of integrin subunits in guinea pig airway epithelial cells. *J Biol Chem* 1992;267:17409–17414.
- Organ LA, Duggan AR, Oballa E, Taggart SC, Simpson JK, Kang'ombe AR, et al. Biomarkers of collagen synthesis predict progression in the PROFILE idiopathic pulmonary fibrosis cohort. Respir Res 2019;20:148.